Author's response to reviews

Title: TIPIT: A randomised controlled trial of thyroxine in preterm infants under 28 weeks gestation: Magnetic Resonance Imaging and Magnetic Resonance Angiography protocol

Authors:

Sze M Ng (M.Ng@liverpool.ac.uk)
Mark A Turner (mark.turner@liverpool.ac.uk)
Carrol Gamble (C.gamble@liverpool.ac.uk)
Mohammed Didi (mohammed.didi@rlc.nhs.uk)
Suresh Victor (Suresh.victor@manchester.ac.uk)
Christina Malamateniou (Christina.Malamateniou@liverpool.ac.uk)
Laura M Parkes (laupar@liverpool.ac.uk)
Anna Tietze (anna.tietze@rlc.nhs.uk)
Lloyd Gregory (lloyd.gregory@manchester.ac.uk)
Vanessa Sluming (vanessa.sluming@liverpool.ac.uk)
Laurence Abernethy (laurence.abernethy@rlc.nhs.uk)
Alan M Weindling (a.m.weindling@liverpool.ac.uk)

Version: 9 Date: 30 June 2008

Author's response to reviews:

Typographical errors were noted on page 1, page 5, page 8 and page 10.